Author's response to reviews

Title: Levels and Patterns of Teenage Marital Pregnancy among Bangladeshi Women

Authors:

Amir M Sayem Mr. (sayemphps@yahoo.com)
Abu Taher MS Nury Mr (nury1974@yahoo.com)

Version: 3 Date: 4 May 2011

Author's response to reviews: see over
Responses to Reviewer

In response to Major Compulsory Revisions, we included the following sentences in the revised manuscript:

We considered women aged 15-29 years because they are the most fertility contributing women in Bangladesh, whereas women aged >29 years have much lower fertility rates [15].

We excluded 42.3% of the participants because they are the most fertility contributing women in Bangladesh, whereas women aged >29 years have much lower fertility rates [15] and because as inclusion of women aged >29 years could produce relatively higher level of recall bias due to relatively longer past event.

In response to Minor Essential Revisions, we have included the following figures in the revised manuscript:

Mostly participants had exposure to television (80.2%), followed by radio (25.4%) and newspaper (16.7%).

In addition, we revised the manuscript through out for editing. Blue colored words/sentences indicate that these are included to the original manuscript while red colored words/sentences indicate that these are omitted from the original manuscript.